Author’s response to reviews

Title: A novel electron transfer flavoprotein dehydrogenase (ETF DH) gene mutation identified in a newborn with glutaric acidemia type II: A case report of a Chinese family

Authors:

Mingcai Ou (scxszx@163.com)
Lin Zhu (zhulin@biosan.cn)
Yong Zhang (z10000y@126.com)
Yaguo Zhang (zzgg0814@126.com)
Jingyao Zhou (854216648@qq.com)
Yu Zhang (517913108@qq.com)
Xuelian Chen (17470430@qq.com)
Lijuan Yang (731658457@qq.com)
Ting Li (757960973@qq.com)
Xingyue Su (1623078983@qq.com)
Qi Hu (109118386@qq.com)
Wenjun Wang (wangwenjun@biosan.cn)

Version: 2 Date: 08 Mar 2020

Author’s response to reviews:

Dear editor,

Thank you for your letter and for revising and improving our paper (ID:MGTC-D-19-00138R1), we have studied your comments carefully and have made revision. The responds to the revision are as follows:

-- Please ensure only one corresponding author has been indicated on the title page.

Response: The corresponding author on Editorial Manager is Wenjun Wang, but we have consulted if it is permissible to have two corresponding authors in different institutions of this paper before we submitted and received a positive reply. So
we except the two corresponding authors Qi Hu (109118386@qq.com) and Wenjun Wang (wangwenjun@biosan.cn) can be indicated on the article.

--Please confirm in the "Consent for publication" section, whether a written consent was obtained from the family/legal guardians on behalf of the patient.

Response: A copy of the written consent is supplied on Editorial Manager.

-- We note that two authors have the same initials (YZ). To distinguish between them in the Authors’ Contributions section please designate them as YZ1 and YZ2, with YZ1 corresponding to Yong Zhang, chosen because this name is furthest up the author list.

Response: We have made correction according to the comments.

-- Please only include one set of figures/tables with your manuscript.

Response: We have made correction according to the comments.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We have made correction according to the comments.

Special thanks to you for your good comments.

Sincerely yours,

Wenjun Wang